Author’s response to reviews

Title: Mutations in the PIGW gene associated with hyperphosphatasia and mental retardation syndrome: a case report

Authors:

Lina Fu (fulina1989927@163.com)
Yan Liu (lyan3022@163.com)
Yu Chen (1030756510@qq.com)
Yi Yuan (yuanyimed@163.com)
Wei Wei (wei.wei@kangso.net)

Version: 7 Date: 23 Jan 2019

Author’s response to reviews:

1. Please include a Funding subsection in the Declarations of your manuscript. If no funding was received for your study you may write “Not applicable” in this subsection.

Reply: We have included a Funding subsection in the Declarations of our manuscript in line 18, page 6.

2. As per BMC submission guidelines for Case Reports, please revise the section headings in: The Abstract so as to read as ‘Background’, ‘Case Presentation’, and ‘Conclusions.’ The Main text so as to read as ‘Background’, ‘Case Presentation’ and ‘Discussion and Conclusions.’ Please do note that in the Main text the ‘Discussion and Conclusions’ is to be one section and heading, not two separate ones.

Reply: In the Main text the ‘Discussion and Conclusions’ has been to be one section and heading, not two separate ones in the Discussion and Conclusions section in line 31, page 4.

3. It is noticed that the patient’s family also underwent DNA analysis. In the Ethics approval and consent to participate subsection of the Declarations please detail the family of the patient’s consent to participate in these methods. Please declare whether consent was provided in written or verbal form.
Reply: We have detailed the family of the patient’s consent to participate in the methods. And we also declared whether consent was provided in written or verbal form in Declarations section in line 21, page 6.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply: We have uploaded our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant figures also are clean versions. Figures have been uploaded as separate files.